Author’s response to reviews

Title: A Plausible Role for Actin Gamma Smooth Muscle 2 (ACTG2) in Small Intestinal Neuroendocrine Tumorigenesis

Authors:

Katarina Edfeldt (katarina.edfeldt@surgsci.uu.se)
Per Hellman (Per.Hellman@surgsci.uu.se)
Gunnar Westin (Gunnar.Westin@surgsci.uu.se)
Peter Stålberg (Peter.Stalberg@surgsci.uu.se)

Version: 4 Date: 18 Mar 2016

Author’s response to reviews:

Replay to reviewer:

After the main section of body of the manuscript "Declations" has been added including:

- Ethics and Consent to Participate statement
- Consent to Publish statement
- Competing Interest statement
- Authors contributions
- Availability of data and materials statement

Best regards

Katarina Edfeldt